Author's response to reviews

Title: War-related psychological sequelae among emergency department patients in the former Republic of Yugoslavia

Authors:

Brett D Nelson BA (bnelso1@jhmi.edu)
Dr William G Fernandez MD, MPH (william.fernandez@bmc.org)
Dr Sandro Galea MD, DrPH (sgalea@nyam.org)
Sarah Sisco MPH, MSSW (ssisco@nyam.org)
Kerry Dierberg BS (kderbe1@jhmi.edu)
Gordana SUBARIC Gorgieva MD, PhD (subariogorgieva@yahoo.com)
Arijit K Nandi BS (akn2005@columbia.edu)
Jennifer Ahern MPH (jahern@nyam.org)
Dr Mihajlo Mitrovic MD (gvozden@eunet.yu)
Michael VanRooyen MD, MPH (mvanrooy@jhmi.edu)
David Vlahov PhD (dvlahov@nyam.org)

Version: 7  Date: 26 May 2004

Iratxe Puebla
Assistant Editor
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Re: MS: 7642205492459296
War-related psychological sequelae among emergency department patients in the former Republic of Yugoslavia

Ms. Puebla,

Just wanted to inform you of two minor changes that have been made:

Submission page- We have removed the qualifications from the authors' list to avoid these being incorporated into the citation of the article.

Introduction- This section has been renamed as "Background"

Thanks for your help. Look forward to seeing this in print, as well as working with you in the future!

--Bill Fernandez